 
CENWP-OP-B                                                                                                                   27 March 2001

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 12 of 2001, which covers the period 18 through 22 March.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 100.4 to 148.5kcfs.  Daily spill averaged 0 kcfs.  Daily project tailwater elevation averaged 10.3’ to 12.5 msl.  Daily project forebay elevation averaged 73.0’ to 74.9’ msl.  River temperature averaged 46 to 48° F.  Spill bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, remain open 4”

1.2 Unit 3 remained out of service for turbine rehab.   Unit 15 remained out of service for gate slot modifications.  

1.3 Powerhouse Two fish units remain in service.  They provide attraction flow for the Washington Shore fish ladder.  The units are being placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.

2.  ACTIVITIES. 
2.1 AWS Closures.   FV4-3 and FV4-4 at B-branch and FV5-3 and FV5-4 are all operating in manual until the computer system is reinstalled.  FV4-3 and FV4-4 were closed for cleaning from 0730 until 1136 and   FV3-9 was closed for cleaning from 1530 until 1534 on 22 March

2.2 STS Inspections.  No STS, ESBS, or VBS inspections this week.  

2.3 Dewatering and Fish Salvage.   No dewatering or fish salvage activities this week.

 
3.   RESEARCH. 

3.1 NMFS Lamprey Tagging and Radio Tracking.  NMFS still track the lamprey back through the project in cooperation with the University of Idaho.

3.2 WDFW Fish Counting.  WDFW continues video counting at Bradford Island and the Washington Shore ladders.  Visual counts are scheduled to begin on 1 April.

3.3 NMFS Saltwater rearing.  The Saltwater rearing facility is under construction.  Collection is expected to start in April.

3.4 PSMFC Adult PIT.  Installation is complete at the Washington Shore ladder.   Collection and tagging of fish is expected to begin April 10th. 

3.5 PSMFC Smolt Monitoring Program.  Smolt monitoring continues at the monitoring facility.  Monitoring at DSM1 is scheduled to begin 15 April.

3.6 WES Hydroacoustic Monitoring.  Transducers are installed at Units 1-5 and Units 7-10.

3.7 USGS Juvenile Radiotracking.  Antennas are installed on the STS’s.  Other antenna installations are expected to be completed the week of 26 March.

3.8 USGS Adult EMG tracking.  Underwater and aerial antennas are installed around Bradford Island and on the pier noses of the spillway.  Collection and tagging is expected to begin next week.

3.9 UI Adult Fallback. UI Adult Radiotracking/Tagging.  Electrical contractor will try to get researcher wiring up to code before 1 April.

4.  FISHWAYS. 
4.1 Project biologists inspected on 19 and 23 of March this reporting week.

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE

All week.
PH1and PH2 collection channel electronic velocity meter.
Meter was not working and should register 1.5-4.0 fps.
Unknown.
Waiting on new velocity meters.

3/23
A-Branch water levels.
Water elevation at the weir reading 1.2’
Unknown
Operators to check controlling valves.

3.23
FG3-25
Should be open at TW elevation 13-34’.
B-Branch is in manual until the computer system is reinstalled.
Operators opened FG3-25.

One day this week.
Bonneville Project
Biologists failed to inspect three times a week.
Workload did not match staffing.


4.3  Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.

Date
Wash. Shore
Bradford Is.

3/18
6.0
5.9

3/19
6.2
6.2

3/20
6.4
6.3

3/21
6.5
6.5

3/22
6.8
6.7

3/23
7.0
6.9

3/24
7.1
7.0

4.4 Sensor Calibration.  Water level sensors were checked for calibration on 19 March this reporting week.  

4.5 Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.

5. MISCELLANEOUS

5.1 The Ice and Trash Chute at the Second Powerhouse was closed at 2120 on 19 March.  
5.2 The high outfall hydrocannons for the Second Powerhouse monitoring facility remain in service.
5.3 As of 22 March, DSM2 orifices 11A-S, F2A-N, and F2B-N are kept open to keep the channel elevation within criteria while Unit 15 is out of service.
5.4 The Powerhouse One Ice and Trash Sluiceway was closed from 1105 until 1612 on 20 March and from 1225 until 1635 on 21 March for USGS BRZ work.
                                                                                                                DEBORAH L. CHENOWETH

                                 

                                                              Operations Project Manager
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